% Grades
{\bf Grades:} All grades will be assigned by the standard 10-point scale.

% +/- table
\begin{center}
\begin{tabular}{c | c |  c | c}
Letter Grade & $-$ & & $+$\\
\hline
% A range
A &
90-91 &
92-97 & 
98-100\\
% B range
B&
80-81 &
82-87 & 
88-89 \\
% C range
C & 
70-71 &
72-77 &
78-79 \\
% D range
D & 
60-61 & 
62-67 & 
68-69 \\
F & 
$< 60$ 
\end{tabular}
\end{center}

The breakdown of your final grade in the course will be as follows:

% breakdown of final grades
\begin{center}
\begin{tabular}{l l}
Homework & $20\%$\\
Midterm 1 & $25\%$\\
Midterm 2 & $25\%$\\
Final Exam & $30\%$
\end{tabular}
\end{center}

